Author’s response to reviews

Title: The effect of methamphetamine abuse on dental caries and periodontal diseases in an Eastern China city

Authors:

Tao Ye (yetaofmmu@foxmail.com)
Dongliang Sun (sdlfmmu@163.com)
Guangying Dong (donggy@fmmu.edu.cn)
Guangjie Xu (guangjiexu@126.com)
Ligang Wang (wangligang8611@sina.com)
Jinjin Du (308594618@qq.com)
Pengcheng Ren (tdrpch@fmmu.edu.cn)
Shibin Yu (yushibin@fmmu.edu.cn)

Version: 6 Date: 14 Dec 2017

Author’s response to reviews:

Response to Editor’s Comments

[Point 1]: Author Emails

We have noted that the email address for the author Tao Ye differs between the title page and the submission system. Please ensure these addresses are edited to be the same. Also, we have been unable to match the email address provided for author Ligang Wang. Please provide an alternative/institutional email address for this author.

[Response]: The author Tao Ye has two email addresses (yetaofmmu@foxmail.com, 461381831@qq.com). We have changed the email address in submission system. We have provided an alternative email address (wangligang8611@sina.com) for the author Ligang Wang.

[Point 2]: Title Page

Please remove the running title, word count and Declaration of the Conflict of Interest sections from the title page.
[Response]: Thank you for your suggestion! We have removed the related information from the title page. Please see the revised manuscript Page 1.

[Point 3]: Declarations Heading

Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

[Response]: According to your suggestion, we have rearranged the related content. Please see the revised manuscript Page 14 Line 10-Page 15 Line 22.

[Point 4]: Oral Consent

Please clarify within the ethics section whether the method of oral consent was approved by the ethical review committee.

[Response]: Thank you for your suggestion. We have added the related information within the section "Ethics approval and consent to participate ", including the ethics protocol number (Protocol No. IRB-REV-2014011). Please see the revised manuscript Page 14 Line 13,18,19.

[Point 5]: Author Details

Please remove the author details section from the manuscript.

[Response]: According to your suggestion, we have deleted the "Author details" section.

[Point 6]: Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All
relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

[Response]: We have done according to your suggestion.